Accurate Prediction of Inducible Transcription Factor Binding Intensities In Vivo

Michael J. Guertin¹⁹, André L. Martins²⁹, Adam Siepel², John T. Lis¹*

1 Department of Molecular Biology and Genetics, Cornell University, Ithaca, New York, United States of America, 2 Department of Biological Statistics and Computational Biology, Cornell University, Ithaca, New York, United States of America

Abstract

DNA sequence and local chromatin landscape act jointly to determine transcription factor (TF) binding intensity profiles. To disentangle these influences, we developed an experimental approach, called protein/DNA binding followed by highthroughput sequencing (PB-seq), that allows the binding energy landscape to be characterized genome-wide in the absence of chromatin. We applied our methods to the Drosophila Heat Shock Factor (HSF), which inducibly binds a target DNA sequence element (HSE) following heat shock stress. PB-seq involves incubating sheared naked genomic DNA with recombinant HSF, partitioning the HSF-bound and HSF-free DNA, and then detecting HSF-bound DNA by high-throughput sequencing. We compared PB-seq binding profiles with ones observed in vivo by ChIP-seq and developed statistical models to predict the observed departures from idealized binding patterns based on covariates describing the local chromatin environment. We found that DNase I hypersensitivity and tetra-acetylation of H4 were the most influential covariates in predicting changes in HSF binding affinity. We also investigated the extent to which DNA accessibility, as measured by digital DNase I footprinting data, could be predicted from MNase-seq data and the ChIP-chip profiles for many histone modifications and TFs, and found GAGA element associated factor (GAF), tetra-acetylation of H4, and H4K16 acetylation to be the most predictive covariates. Lastly, we generated an unbiased model of HSF binding sequences, which revealed distinct biophysical properties of the HSF/HSE interaction and a previously unrecognized substructure within the HSE. These findings provide new insights into the interplay between the genomic sequence and the chromatin landscape in determining transcription factor binding intensity.

Citation: Guertin MJ, Martins AL, Siepel A, Lis JT (2012) Accurate Prediction of Inducible Transcription Factor Binding Intensities In Vivo. PLoS Genet 8(3): e1002610. doi:10.1371/journal.pgen.1002610

Editor: Michael Snyder, Stanford University School of Medicine, United States of America

Received October 11, 2011; Accepted February 6, 2012; Published March 29, 2012

Copyright: © 2012 Guertin et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was funded by a National Institutes of Health grant (GM25232) to JTL. A supplement (GM25232-S1) supported ALM and AS and a predoctoral training grant (T32-GM007617) supported MJG. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: jtl10@cornell.edu

• These authors contributed equally to this work.

Introduction

Binding of transcription factors (TFs) to DNA elements is necessary to establish and maintain functional changes in gene expression levels. The mechanism by which these factors seek out and bind to their cognate motif elements remains an area of active investigation (reviewed in [1]). TFs are present at cellular concentrations that allow binding to sites that are degenerate from the consensus sequences, and genomes of eukaryotes are littered with potential degenerate binding sites; however, only a small fraction of potential binding sites are recognized in vivo. Moreover, TF binding sites vary dependent upon cell type and cellular conditions. In vivo, TF binding is potentially dependent upon motif accessibility and the surrounding chromatin landscape. Therefore, determining a comprehensive set of potential genomic binding sites and quantifying the joint effects of DNA sequence and chromatin landscape upon binding intensity remains a challenge.

Experimental approaches to characterize TF binding sites include assays such as ChIP-seq, protein binding microarrays (PBM) [2], iterative rounds of protein-DNA binding and selection with a complex oligonucleotide library [3], or extrapolation from DNase I hypersensitivity regions [4]. However, perhaps the most direct way to determine all potential TF binding sites within a genome is to incubate purified TF and naked sheared genomic DNA in vitro, and then specifically quantify the TF-bound DNA [5]. This in vitro method allows binding sites to be interrogated in their native sequence context without the confounding effects of chromatin and cooperation between chromatin-bound factors.

It is challenging to predict in vivo TF binding accurately even when all potential in vitro binding sites have been characterized, because the chromatin landscape dramatically influences binding and it changes dynamically with development and with alterations in cellular nutrition and environment [6,7]. Recent TF binding site modeling efforts have considered genomic nucleosome occupancy or DNase I hypersensitivity data to account for the effect chromatin has upon in vivo TF occupancy [8–11]. However, these models are limited in that they rely upon genomic accessibility data and TF binding data produced under the same conditions. To date there are no data sets that describe the full set of potential TF binding sites, the chromatin state data prior to binding, and occupied binding sites in vivo, in a single inducible system. Integration of these three data sets would allow one to

Author Summary

Transcription factors (TFs) bind DNA to modulate levels of gene expression. TF binding sites change throughout development, in response to environmental stimuli, and different tissues have distinct TF binding profiles. The mechanism by which TFs discriminate between binding sites in a context dependent manner is an area of active research, but it is clear that the chromatin environment in which potential binding sites reside strongly influences binding. This study used the Heat Shock TF (HSF) to study the effect chromatin has upon induced HSF binding. We implemented an experimental technique to quantify all potential HSF binding sites in the genome. These data were incorporated into a modeling framework along with chromatin landscape information prior to HSF binding to accurately predict the intensities of inducible HSF binding sites. DNase I hypersensitivity and tetra-acetylation of H4 were the most influential covariates in the model. The binding data enabled the development of a more complete HSF/DNA interaction model, providing insight into the biophysical interaction of HSF trimer subunits and target DNA pentamers.

decouple the effect TF binding has upon chromatin state from the effect pre-existing chromatin state has upon induced TF binding.

The heat shock response of *Drosophila* is a model system extensively used to study the general functions of sequence specific activators and how they function to regulate transcription (reviewed in [12]). The master regulator of the heat shock genes, Heat Shock Factor (HSF), has a modest affinity for DNA under non-stress conditions [6,13,14], and upon stress, HSF homotrimerizes and inducibly binds to a conserved consensus motif at over 400 sites in the *Drosophila* genome [6,14]. While over 95% of the HSF binding sites contain an underlying HSF sequence motif element (HSE), the vast majority of predicted genomic HSEs remain HSF–free following heat shock. Therefore, the chromatin landscape most likely plays a prominent role in determining binding of HSF.

Here, we describe an experimental technique, protein/DNA binding followed by high-throughput sequencing (PB–seq), to quantify the binding potential of all binding sites within a genome. We then develop a quantitative model that incorporates HSF PB–seq data, together with HSF ChIP-seq in *Drosophila* S2 cells [6] and S2 cell chromatin data, that accurately predicts observed in vivo HSF binding profiles. Moreover, our model allows us to quantify the relative importance of the chromatin features influencing HSF binding intensity. Finally, we develop a sequence model that uses HSF PB–seq data to characterizes the relationship between positions within the HSE and provide biophysical insight into the mechanisms by which HSF interacts with its cognate element.

Results

Quantification of the absolute binding affinity of all genomic *Drosophila* HSEs

We performed an in vitro binding experiment with purified HSF (Figure S1) and naked, sheared genomic *Drosophila* DNA, to derive an accurate set of potential HSF binding sites in the *Drosophila* genome. HSF–bound DNA was specifically eluted and detected by high throughput sequencing (Methods). The HSF PB–seq experiment yielded 68% of the sequence tags within peaks. In contrast, typical ChIP-seq protocols are more inefficient and the majority of DNA (60% to >99%) sequenced is uninformative background DNA [15].

Peak calling revealed 3952 HSF–binding peaks (p<0.01; 2848 peaks were common to both experimental replicates), which include 60% of the previously identified high-confidence HSF binding peaks in vivo [6]. The naïve expectation is that every in vivo HSF peak should have a corresponding in vitro peak, but it is not surprising to observe an incomplete overlap of in vivo by in vitro peaks, for various reasons. As will be discussed, binding sites detected in vivo but not in vitro tend to be more degenerate and have higher DNase I accessibility. Additionally, in vivo binding sites that are dependent upon cooperative interactions with prebound chromatin factors, long range DNA interactions, post-translational modifications of HSF [16], higher-order chromatin structure, or bridging protein interactions [17] will not be detected in the current form of PB–seq.

Underlying the in vitro binding peaks, we detected 3735 clusters of HSF binding site HSE sequences (2896 in peaks common to both replicates) at 20% HSE False Discovery Rate (FDR). We used clusters of co-occurring sites due to the uncertainty in HSE detection (see Methods). Furthermore, the majority, 3389 clusters (2586 in peaks common to both replicates) are not detectably bound in S2 cells in vivo. Figure 1 shows two examples of in vitro binding sites flanking the Cpr67B gene that are not bound in vivo. Moreover, the in vitro binding data quantifies differences in the in vitro and in vivo HSF binding intensity, such as the peaks within each of the promoters for Hsp23 and Hsp26 (Figure 1).

The PB–seq experiment allows for an estimate of the relative binding intensity of each HSE, based on the number of sequence tags associated with it. To compute the dissociation constant (Kd) values it is necessary to have estimates for both the fraction of bound and free HSE in the PB–seq experiment. Since the PB–seq data only provides information on the bound fraction, we needed to determine the absolute Kds for two HSEs that are found within the PB–seq data in order to provide enough information to estimate the free fraction (see Methods).

To generate the HSF/HSE Kd measurements, we performed electrophoretic mobility shift assays (EMSA). The EMSAs were performed with purified HSF and HSEs that are only modestly degenerate from the consensus. We found that HSF binds to the first HSE with ~42.6 pM interval: 36.8–49.4 pM; Figure 2A and 2C) and the second HSE with ~224 pM affinity (95% confidence interval: 181–276 pM; Figure 2B and 2D). The resulting two absolute Kd values enabled us to transform PB–seq read depths into absolute Kd values (Figure 2E and Methods). We confirmed the transformation of the relative Kd values to absolute Kds by performing band shifts with genomic HSEs of different predicted Kd values (Figure S2). The experimental verifications of the measurements are within the estimated error of the EMSA confidence interval and the variability between PB–seq replicates (Figure S3).

Taken together, these measurements allow us to characterize the binding energy landscape for HSF across the entire *Drosophila* genome, in the absence of chromatin. Our estimated Kd values for isolated HSEs in the *Drosophila* genome ranged from 40–400 pM (Figure 2E). These in vitro binding results demonstrate the feasibility and efficiency of combining high-throughput detection methods with classic EMSA and competition experiments to quantify the binding energy for the comprehensive set of potential genomic binding sites for a sequence-specific TF.

Chromatin features and PB-seq data predict HSF binding intensity in vivo

Our data reveals substantial differences between in vivo and in vitro binding intensities (Figure 3A), underscoring the role of chromatin in determining in vivo binding site selection and



Figure 1. In vitro binding reveals potential HSF binding sites. The blue box highlights strong differences in the usage of potential binding sites in vivo at the Cpr67B locus, while the green boxes highlight differences in the magnitude of binding to major heat shock genes promoters, despite comparable in vitro binding affinities. doi:10.1371/journal.pgen.1002610.g001

affinity. We found DNase I hypersensitivity was the most important predictor of HSF binding; therefore, we scaled the in vivo and the in vitro read counts so that they were approximately equal at in vivo sites with high DNA accessibility (Methods, Figure S4). After this normalization, we partitioned the binding sites that were detectable in vitro into classes: "unaffected" sites, bound at



Figure 2. Recombinant HSF binds HSEs with picomolar affinity in vitro. A and B) The mobility of the constant 200 attomole HSE probe shifts into a trimeric-HSF:HSE complex as increasing HSF is added. There is no HSF in the left-most lane, the right-most lane contains 3 nM HSF (1 nM trimeric HSF), and the intervening lanes contain two-fold serial dilutions of HSF. C) A hyperbolic curve based on the Kd equation (see Methods) was modeled using the band shift data, indicating a Kd of 42.6 pM (95% confidence interval of 36.8–49.4 pM). D) A hyperbolic curve based on the Kd equation (see Methods) was modeled using the band shift data, indicating a Kd of 224 pM (95% confidence interval of 181–276 pM). E) The intensity of each isolated HSE in the *Drosophila* genome is transformed to an absolute Kd using the absolute Kds calculated from band shift data in panels A and B. The Kd values range from 40–400 pM. doi:10.1371/journal.pgen.1002610.q002



in vitro Intensity

in vivo Intensity Relative to in vitro

Figure 3. In vitro and in vivo binding of HSF to genomic HSEs do not correlate. A) A scatter plot comparing the observed in vivo HSF binding intensity and in vitro binding intensity for each isolated HSE indicates that the vast majority of in vivo binding is suppressed (green) or abolished (blue), if we assume that the top seven most DNase I hypersensitive isolated HSE clusters provide the best estimates for sites that are minimally influenced by chromatin. After scaling, red points have similar in vivo and in vitro intensity, black points may be enhanced in vivo, while green and blue points are suppressed and abolished, respectively. B) The points from panel A were categorized, and the resulting bar chart shows the relative frequencies of each category. doi:10.1371/journal.pgen.1002610.g003

comparable affinities in vivo and in vitro (55 red points in Figure 3A; 2% of all sites); "suppressed" sites, with reduced, but detectable, in vivo intensity (365 green points; 13%); and "abolished" sites, below the in vivo threshold for detection (2223 blue points; 76%). In addition, sites not detectable in vivo or in vitro were labeled "background" (249 gray points; 9%), and sites with stronger relative in vivo intensity compared to in vitro were labeled "enhanced" (4 black points; 0.1%).

PB-seq data reveals potential HSF binding sites, providing the opportunity to model the effect that non-stressed chromatin landscape has upon induced HSF binding intensity. There is a wealth of chromatin data available for S2 cells during unstressed conditions [18,19], and heat-shock induced binding sites of HSF in S2 cells are also known [6]. We used DNase I hypersensitivity data [18], MNase data [19] and ChIP-chip data for 9 factors and 21 histone modifications for unstressed Drosophila S2 cells (Table S1) [18] to predict the intensity of inducibly bound in vivo HSF-bound sites (Figure 4A, Figure S5 and Figure S6). For our statistical model, we selected a *rules ensemble* [20], a linear regression model in which some terms are combinations of covariates known as "rules". This approach allowed us to capture fairly complex interactions between covariates. For example, a rule might apply when H3K27ac and DNase I hypersensitivity both exceeded designated thresholds (value ranges can also be expressed). Each rule's coefficient is added to the predicted value if, and only if, the rule applies. When there is only one covariate, the model reduces to a linear regression.

The Pearson's correlation coefficient between HSF ChIP-seq data for the model incorporating all these data sets was r = 0.62 (Figure S6 and Figure S7). As the large number of covariates

brings with it some danger of overfitting, we tested combinations of the four classes of covariates: DNase I hypersensitivity, MNase, histone modifications/variants, and non-histone factors (Figure 4B, Figure S6, Figure S7). Of notice, the correlation of the linear regression model that incorporates DNase I data was r = 0.64 on the test data (Figure 4B and Figure S7B). Our study is consistent with a previous study that obtained r = 0.65 for actual and inferred TF binding intensities using a DNase I dependent model [8].

Other covariate classes produce similar, but lower, correlations. The rules model using histone modifications and histone variants yielded r = 0.57 (Figure 4B and Figure S7), while a rules model incorporating non-histone bound chromatin factors yielded r = 0.58 (Figure 4B and Figure S7). Combining covariate classes further improves the correlation to as much as r = 0.70 (Figure S6 and Figure S7). We also examined the Receiver Operator Curves (ROC) for the different covariate combinations (Figure S8) and found concordant results. If we assume that the PB-seq, genomic ChIP, DNase I-seq, and MNase-seq experiments are maximally resolved and sensitive, with no experimental noise, an approximate upper bound is given by r = 0.90, as observed for two HSF-ChIP-seq replicates [6]. Notably, the higher resolution of the DNase I-seq data, compared to the ChIP-chip data, may be why DNase I-seq alone is strongly predictive in the linear regression model and most influential in the rules ensemble models. Notably, we used the chromatin landscape prior to induced TF binding to predict binding intensity, whereas previous models have used the chromatin landscape present when the TF is bound in order to infer binding intensity [8] or infer binary binding events [10,11] (see Discussion).



Figure 4. Genomic chromatin and PB-seq data accurately predict in vivo HSF binding intensity. A) The intensity of in vivo ChIP-seq peaks is not recapitulated by in vitro PB-seq data; however, genomic DNase I hypersensitivity data and histone modification ChIP-chip data can be used to accurately predict HSF binding intensity. B) The experimentally determined ratio between in vivo ChIP-seq HSF intensity and in vitro PB-seq intensity is plotted against the predicted in vivo/actual PB-seq ratio. The Pearson correlation for each model is shown. doi:10.1371/journal.pgen.1002610.g004

Our data and modeling indicated that the presence of active chromatin features, such as histone acetylation and DNase I hypersensitivity, had a significant influence on the predictive power of the model, while repressive features had minimal influence (Figure S9). DNase I hypersensitivity was a strongly predictive covariate in the model when used in a simple linear regression model (Figure 4), or in combination with histone modification and non-histone factor covariates in the rules (Figure S9E-S9G, S9J, S9K, and S9M). Tetra acetylation of H4 and H3K9ac were the most informative histone marks in the model that used histone variants and histone modifications as covariates (Figure 5A). GAGA associated factor (GAF), which has a proposed role in permitting HSF binding [21], was the most influential factor in the HSF binding prediction model that considered all chromatin-binding factors (Figure 5B).

Defining genome-wide DNA accessibility by chromatin composition

The analysis above indicates that DNA accessibility, as measured by DNase I hypersensitivity, is a primary determinant of binding intensity. Previous studies have similarly shown that TF binding sites correlate strongly with DNase I hypersensitive sites [8,10,11,22]. For instance, histone acetylation causes local chromatin decondensation by reducing the ionic interactions between lysine residues and DNA and promotes accessibility, but the extent to which combinations of histone marks and TFs act together to dictate chromatin accessibility is not known. Therefore, it is of interest to see whether DNA accessibility can be predicted from specific features of the chromatin landscape, such as histone modifications and non-histone chromatin bound factors. In addition, accurate predictions of DNA accessibility would be of practical use, because direct measurements are often not available.

To address this question, we applied our rules ensemble framework to predict DNase I hypersensitivity (the best available proxy for DNA accessibility) from ChIP-chip data for histone features, non-histone chromatin bound factors, MNase data and combinations of these covariate pools (Figure 6). Tetra-acetylation of H4 and H3K9 acetylation were most influential in the model that uses histone modifications, bulk histone and histone variant intensities (Figure S10E); the correlation coefficient for this model using the test data is 0.51 (Figure S11B). The model that uses non-histone factor ChIP-chip data obtains a correlation of 0.52 (Figure S11B), which is consistent with TFs having characteristic DNase I hypersensitivity footprints [10,11]. The model that combines both histone data and non-histone data into a rules model performs the best on the test set, with a correlation of 0.60 (Figure S11B). Repressive histone marks appear to contribute little to generating the DNase I hypersensitivity pattern (Figure S10) and the lack of active chromatin marks appears to be sufficient to package DNA into inaccessible units. These models reinforces the notion that the biochemical composition of chromatin permits DNase I hypersensitivity and quantifies the contributions individual modifications, and combinations thereof, make to DNase I hypersensitivity (Figure S11). As more and higher-resolution genome-wide data becomes available, these models will be refined.

Dissection of the Heat Shock Element

PB-seq provides the opportunity to model the sequencedependent binding preferences of a purified TF genome-wide and independent of chromatin or other factors. In the case of HSF, the consensus binding site is well characterized and consists of three pentamers, ÒAGAAN NTTCT AGAANÓ, (here denoted pA, pB, and pC), each bound by a monomer of the HSF homotrimer. Note that the consensus sequences for pA and pC are identical, while the one for pB is their reverse complement. Of course, the consensus HSE is a crude summary that ignores subtleties in the base preferences at each position. A positionspecific scoring matrix (PSSM) provides a somewhat improved description but still ignores dependencies between positions within the binding site. We sought to use genome-wide binding sites from PB-seq to produce an improved model for the sequence preferences at HSEs.

We began by computing the mutual information for all pairs of HSE positions based on the identified in vitro binding sites. We found negligible evidence of correlated base preferences between positions, but we did observe that some pentamers within PB–seq peaks adhered closely to the consensus motif while others did not. This led us to formulate a probabilistic model that allows each pentamer in an HSE to closely match the consensus ("strict") or diverge from it more substantially ("relaxed"), and considers all possible combinations of pentamer composition (Figure S12). More specifically, we described each of the three pentamers using a two-component mixture model, with a latent variable indicating "strict" or "relaxed" binding preferences, and estimated the joint distribution of these three latent variables from the data.

The model parameters—the position-specific nucleotide probabilities and prior distribution for the combinations of strict/relaxed pentamers—were estimated from the data by maximum likelihood using an expectation maximization algorithm (see Methods). In fitting the model, we considered only the 1309 isolated HSEs, sequence elements that were at least 200 base pairs away from any other degenerate HSE motif, to avoid complications arising from overlapping HSEs. The model fit the data substantially better than did a simple PSSM (lnL = -15442 vs. lnL = -15673 for the PSSM; Akaike information criterion [AIC] = 15636 vs. AIC = 15763 for the PSSM), suggesting that it effectively captures important dependencies between positions.

Based on the estimated model parameters, we computed a posterior probability distribution over all combinations of pentamer stringency and order for each HSE (Methods; Figure 7B). These values were averaged across HSEs to obtain expected genome-wide fractions of HSEs having each of the strict/ relaxed pentamer combinations. We found that binding sites with strict pB and pC, and relaxed pA, were most frequent (an expected 38% of sites), indicating that this configuration is preferred (Figure 7B). The next most frequent configurations were a relaxed pB flanked by a strict pA and pC (33%), and a strict pA and pB combined with a weak pC (29%). Interestingly, combinations of three strict pentamers occur at negligible frequency. Indeed, only 5 out of 1309 isolated genomic HSEs matched the consensus sequence exactly, while 148 differed from it by a single mismatch. Configurations with at most one strict pentamer were also rare. Together, these results indicate that the biophysical interactions of the pentamers within the binding sites are critically dependent

Histone Mods.



В

Α

non-Histone Factors



Figure 5. Histone acetylation and GAF occupancy are important covariates in predicting HSF binding intensity. Plotted are the relative values of the sums of the coefficients associated with all rules that reference each covariate in the rules ensemble [20]. Results are shown for (A) the histone variant and modification model and (B) the non-Histone factor model. doi:10.1371/journal.pgen.1002610.g005

upon their composition and position relative to the other pentamers in an HSE.

While the three estimated strict pentamer matrices were similar (Figure 7A top), the relaxed matrices showed substantial

differences with respect to each other (Figure 7A bottom). For example, the relaxed pA matrix indicates that 70–80% of HSEs containing a weak pA have the consensus base at positions two, three and four. In contrast, position 12 in pC (the analog of



Figure 6. DNase I hypersensitivity can be inferred using histone marks and MNase data. A) The intensity of DNase I hypersensitivity landscape is inferred by models (colors) that use histone modification profiles, non-histone factor profiles, DNase I data and MNase-seq data. B) The experimentally determined DNase I hypersensitivity data is plotted against inferred intensity for the various models. The Pearson correlation for each model is shown. doi:10.1371/journal.pgen.1002610.g006

PLoS Genetics | www.plosgenetics.org



Figure 7. Pentamers within the HSEs are dependent upon their consensus match and also their position relative to the other pentamers. A) The mixture model defines each pentamer within the HSE as strict or relaxed depending upon how well it conforms to the canonical HSE. Note that the position of relaxed pentamers strongly influences their composition. B) A probabilistic sequence model reveals that the presence of two strict (red) and one relaxed (blue) pentamer provides the best explanation of the data. doi:10.1371/journal.pgen.1002610.g007

position 2 in pA) almost invariably contains a G in all HSEs, while positions 7 and 8 in pB (analogous to positions 3 and 4 in pA) have only modest base preferences in HSEs containing a weak pB. This analysis indicates that each monomeric HSF/pentamer interaction has distinct biophysical properties within the context of the broader HSF/HSE interaction.

We also devised a simplified model, with a single strict matrix shared by all three pentamers, and a single relaxed matrix obtained by applying a "dampening" factor to the strict matrix (Figure S13, Methods). This model further supports the strict/ relaxed pentamer split ($\ln L = -15908$ vs. $\ln L = -16048$ for a single-monomer PSSM; and AIC = 15952 vs. AIC = 16078), although both the full model and the full PSSM fit the data better (lower AIC). Moreover, not only was the simplified model still able to reproduce the posterior distributions over pentamer configurations of the full model, but it was also able to replicate synthetic patterns from simulated data (Figure S14).

Finally, the preference for single pentamer degeneracy was also observed independently by comparing the pentamer-specific KLdivergence in PSSMs obtained from subsamples of HSF bound peaks (Figure S15; Methods).

Discussion

The PB-seq technique combined with EMSA and competition assays provides a straightforward, yet versatile and powerful framework for characterizing all potential binding sites in a genome, regardless of tissue specificity, developmental stage, or environmental conditions. Comparing in vitro and in vivo binding profiles, in the context of pre-induction genomic chromatin landscape, revealed DNase I hypersensitivity, H4 tetra-acetylation, and GAF as critical features that modulate cognate element binding intensity in vivo. Furthermore, DNase I sensitivity was found to be strongly influenced by high GAF occupancy and histone acetylation, while repressive factors were minimally influential in the statistical models. Finally, the full set of potential genomic binding sites provided a rich data set that was used to build more detailed sequence models, which tease apart substructure and features that are lost with traditional PSSM models.

One initially surprising observation from our study was that 40% of the in vivo HSF peaks were not found in vitro. We believe that the limited dynamic range for quantifying in vitro binding affinity may be responsible for the lack of detectable in vitro peaks. Although we quantify in vitro binding over an order of magnitude (40-400 pM), the experimental concentrations of HSF and genomic DNA and our depth of sequencing do not permit the detection of lower affinity HSF binding sites. For instance, only eleven sequence tags would be predicted to underlie a hypothetical 5 nM HSF binding site, and these would not be distinguishable from background. Upon further examination, we find that the composite HSE representing those in vivo binding sites that were not found in vitro is more degenerate than those found using both assays (Figure S16A). Moreover, the in vivo sites that were not found using PB-seq were also more accessible in vivo (Figure S16B), in support of our hypothesis. Performing PB-seq at a range of protein and DNA concentrations, or increasing sequence coverage would expand the dynamic range of quantification by PB-seq.

Other possible explanations for this observation include cooperative interactions with pre-bound chromatin factors, longrange DNA interactions, post-translational modifications of HSF, higher-order chromatin structure, or bridging protein interactions. The influence of DNA modifications and immediate flanking sequence do not contribute to this disparity, since we use large fragments of purified genomic DNA. Bridging protein interactions [17], which do not involve HSF directly binding to DNA, appear not to be responsible for our results because 95% of in vivo peaks encompass at least one HSE near the peak center [6]. However, if other proteins were cooperating with HSF in vivo to enhance HSF binding intensity at low affinity binding sites, then some of these peaks may not be observed in vitro. Since our PB-seq experiment used recombinant HSF in the binding experiments, we would also not capture differences in binding site affinities that are due to post-translational modifications of HSF [16]. To overcome these potential limitations, PB-seq could be adapted to include known bridging/cooperative factors and proteins could be purified from in vivo sources to capture indirect or modification-dependent interactions.

The notion that motif accessibility is driving inducible TF binding in vivo is supported by independent studies of distinct TFs: STAT1, HSF, glucocoticoid receptor (GR), and GATA1 [6,22-24]. These studies show that the chromatin landscape prior to TF binding influences inducible TF binding. In the first study, it was found that a large fraction of STAT1 induced binding sites contained H3K4me1/me3 marks prior to interferon-gamma (IFN- γ) induced STAT1 binding [23]. Our group previously found that inducible HSF binding sites are marked by active chromatin compared to sites that remain HSF-free [6]. A more recent study has shown that inducibly bound GR sites are marked by DNase I hypersensitive chromatin prior to GR binding [22]. Likewise, the permissive chromatin state at GATA1 binding sites is established even in GATA1 knock out cells [24]. While these correlations are instructive, no previous attempt has been made to model inducible TF binding using biological measurements of chromatin landscape present prior to TF binding. Recent models have successfully inferred TF binding profiles using DNA sequence and chromatin landscape data, generated at the same time the TF is bound [8-11]. However, these models do not distinguish between the influence TFs have upon local chromatin and the chromatin features that permit TF binding. In contrast, we modeled the changes between HSF in vitro binding (PB-seq) and in vivo binding (ChIP-seq) landscapes as a function of the non-heat shock chromatin state. This produced a quantitative model describing the important features that modulate the in vivo HSF binding intensity. Moreover, the use of our rules ensemble model enabled the capture of potential interactions between these chromatin features.

Our study reveals that DNase I hypersensitivity and acetylation of H4 and H3K9 are strong predictors of inducible HSF binding intensities, however the molecular events and factors that precede TF occupancy to maintain accessible chromatin remain poorly characterized. For instance, the degree to which pioneering factors or flanking DNA sequence, individually or in combination, maintain or restrict accessibility remains unclear. A recent study highlights the biological consequences of maintaining the inaccessibility of TF binding sites, in order to repress expression of tissuespecific transcription factors in the wrong tissues. The authors found that ectopic expression of CHE-1, a zinc-finger TF that directs ASE neuron differentiation, in non-native C. elegans tissue is not sufficient to induce neuron formation [25]. However, combining ectopic CHE-1 expression with knockdown of lin-53 did modify the expression patterns of CHE-1 target genes in nonnative tissue, effectively converting germ line cells to neuronal cells [25]. LIN-53 has been implicated in recruitment of deacetylases,

and deacetylase inhibitor treatment mimics *lin-53* depletion, suggesting that LIN-53 is actively maintaining CHE-1 target sites inaccessible in germ cells.

Alternatively, functional TF binding sites could be actively maintained in the accessible state. HSF binding within ecdysone genes has a functional role in shutting down their transcription [14], and activating ecdysone-inducible genes containing inaccessible HSEs causes chromatin changes that are sufficient to allow HSF binding [6]. In this special case of HSF–bound ecdysone genes, active transcription and the corresponding histone marks are mediating access to HSEs, in order for HSF to bind and repress transcription upon heat shock. A more recent study has shown that activator protein 1 (AP1) actively maintains chromatin in the accessible state, so that GR can bind to cognate elements [26].

Although TF accessibility to critical genomic sites appears to be actively maintained, many binding sites may be a non-functional result of fortuitous TFBS recognition. It has long been hypothesized that the binding affinities for TF/DNA interactions are sufficiently strong to allow promiscuous binding at the cellular concentrations of TFs and DNA [27,28]. There are roughly 32,000 HSF molecules per tetraploid S2 cell [29] and the dissociation constants for trimeric-HSF/HSE interactions are in the picomolar range (Figure 2E); therefore much of the in vivo HSF binding may be non-functional promiscuous binding. Additional investigation will further illuminate the role of chromatin context in TF binding and the mechanisms by which programmed developmental or environmental chromatin changes permit or deny TF binding.

Elucidating the rules that govern accessibility is essential for predicting in vivo occupancy of TFs. Diverse transcription factors [7], from a broad spectrum of organisms [22], bind their sequences based on site accessibility. We found that chromatin accessibility as measured by DNase I hypersensitivity could be inferred using ChIP-chip data for various histone modifications and transcription factors. Although our model can infer accessibility based on chromatin composition, the mechanism by which accessibility originates is not addressed. Previous studies have shown that activators, such as HSF, glucocorticoid receptor, and androgen receptor bind to their cognate sites and direct a concomitant increase in local acetylation, DNase I hypersensitivity, and nucleosome depletion [6,22,30,31]. Androgen receptor also acts to position flanking nucleosomes marked by H3K4me2 [31]. These post-TF binding chromatin changes that occur are the result of acetyltransferase and nucleosome remodeler recruitment, both of which functionally interact with activators. For instance, both GR and GATA1 interact with the nucleosome remodeling complex Swi/Snf [32,33]. Concomitant increases in locus accessibility likely allow large molecular complexes such as RNA Pol II and coactivators to access the region that in turn can reinforce and maintain active and accessible chromatin.

Thorough biophysical characterization of TF binding site properties is critical for accurate predictions of TF binding, underscoring the need for more complete models of TF binding. While the commonly used PSSM model makes the assumption of base independence, recent work has revealed that richer models providing for interactions between positions are necessary [34,35]. Our model captures critical features of the HSF/HSE interaction that are lost with simpler computational models, namely the interdependencies between the sub-binding sites of each HSF monomer. Consistent with our model, a series of in vitro experiments with *S. cerevisiae*, *D. melanogaster*, *A. thaliana*, *H. sapien* and *D. rerio* HSFs indicate that HSF from each of these species can bind to discontinuous HSEs containing canonical pentamers that contain intervening five base pair gaps [36,37]; interestingly, however, *C. elegans* HSF strictly binds to continuous HSEs that do not contain gaps [36]. The complex interactions between positions within a binding site are a critical aspect of inferring whether a polymorphism or mutation affects TF binding. These features should prove useful in providing degenerate HSE sequences for optimal co-crystallization of trimeric HSF and DNA and inferring changes in DNA sequence that affect HSF binding within and between species.

In conclusion, the data and models presented here reinforce both the importance of chromatin landscape in modulating in vivo TF binding intensity and how genome wide, chromatin free, binding assays contribute to the understanding of TF sequence binding specificity.

Methods

Cloning and purification of recombinant HSF

Drosophila HSF was N-terminally tagged with glutathione stransferase and a tobacco etch virus (TEV) protease cleavage site. The C-terminus of the recombinant HSF was fused to the 3xFLAG epitope. Recombinant HSF was purified from *E. coli* with glutathione resin as previously described [38], with the following modifications: HSF–3xFLAG elution was achieved by addition of 6xHistidine tagged TEV protease and TEV protease was cleared from the HSF preparation using a Nickel-NTA column. Densitometry was used to show that the HSF protein preparation was 40% full length HSF–3xFLAG, and known amounts of bovine serum albumin (BSA) were used to quantify the HSF (Figure S1).

Band shift assay

Serial two-fold dilutions of recombinant HSF, from 3 nM (1.5 nM for the 221 pM HSE) to 23.3 pM, was incubated with 200 attomoles of radiolabeled dsDNA containing modestly degenerate HSEs (chrX:3380775-3380824 (224 pM), chr2L:5009892-500994 (42.7 pM), chr2R:3529792-3529841 (308 pM), chr3L 13470978-13471009 (221 pM), and chr3L:4073542-4073591 (97.5 pM)) and allowed to come to equilibrium for 30 minutes in a total of 10 µl of 1xHSF binding buffer (20 mM HEPES pH 7.9, 10% glycerol, 1 mM EDTA, 4 mM DTT, 3 mM MgCl2, 100 mM NaCl, 0.1% NP-40, and 300 µg/ml BSA) at room temperature. Binding reactions were loaded in a 3% agarose TBE (10 mM Tris-HCl pH 8.0, 25 mM boric acid, and 1 mM EDTA) gel and electrophoresed at 50 Volts for 2 hours. The HSF-bound probe and free probe were quantified by densitometry and the dissociation constant, Kd = ([A][B])/[AB], was estimated using a non-linear least squares method on the function $[AB]/[A]_{total} = [B]/([B]+Kd)$ where $[AB]/[A]_{total}$ is the measured shifted fraction and [B] is the free HSF trimer concentration.

PB-seq: Genomic in vitro binding experiment

We incubated 600 pM HSF and 2500 ng genomic DNA (sonicated to 100–600 bp fragment size as previously described [6]) in 1500 μ l final volume of 1xHSF binding buffer and let it come to equilibrium for an hour at room temperature. We added 20 μ l ANTI-FLAG M2 affinity gel for 10 minutes and washed 8 times with 1xHSF binding buffer to remove unbound DNA, 3xFLAG peptide was added to a final concentration of 200 ng/ μ l to specifically elute HSF and HSF–bound DNA. The mock IP was done in the absence of recombinant HSF. We attribute the in vitro binding assay's low background to the design of the experiment. Since recombinant C-terminally 3xFLAG tagged HSF was used, the HSF–associated DNA could be specifically eluted by the addition of excess 3xFLAG peptide. In contrast, standard ChIP

protocols rely on non-specific elution of all protein and DNA that binds the resin.

Illumina library preparation

The sample preparation was as previously described [6], except that 15 rounds of amplification were performed in this case.

PB-seq HSF peaks and HSE sites

The PB-seq reads were aligned to the *Drosophila* Genome (BDGP R5/dm3) using BWA (v 0.5.8c) [39]. We obtained 5,052,425 uniquely aligned reads for replicate one, 4,694,846 for replicate two and 5,410,049 for the mock. Files that contain raw sequence data and uniquely aligned reads were deposited into NCBI's Gene Expression Omnibus (GEO) [40], accession number GSE32570.

We called peaks using MACS (v 1.3.7.1) [41], both for each individual replicate and for the merged set, using a tag size of 55 bp, a starting bandwidth of 100 bp and an appropriate genome size. After experimenting with several p-value thresholds, we selected a value of p=0.01, which achieved a good tradeoff between maximizing the number of called peaks and ensuring consistency between replicates. Our results were largely unaffected by the 'mfold' parameter (the threshold for fold enrichment relative to background for inclusion in the peak model), so we left this parameter at its default value.

To improve our sensitivity in binding site detection, we made use of an ensemble of position weight matrices (PSSMs), rather than a single matrix. We sampled 10,000 sets of 100 peaks and used the program MEME [42] for motif discovery in each set. As input, MEME was given the 100 bp sequence centered at each peak summit. We used a fixed motif width of 14 bp, a second order background Markov model estimated from the entire peak set, and the 'zoops' model (zero or one site per sequence) with the restriction that at least 75% of the sequences must contain a site. The resulting PSSMs were compared by KL-divergence against the canonical monomer PSSM (four base pair unit with consensus AGAA) estimated from the previously published in vivo highconfidence HSF binding sites detected by ChIP-seq [6]. In each PSSM, one of the three monomers had on average about twice the KL-divergence as the other two. Figure S15 shows a scatter plot of the KL-divergence of the PSSMs in the ensemble

Each peak was scanned for matches to all PSSMs in the ensemble, allowing for overlapping sites. The score at each position was taken to be the maximum score across the ensemble. Peaks were split into three groups by GC% quantile, and for each group a 10 kbp sequence was simulated from a second order Markov model, which was then used to estimate the FDR associated with the score.

In our context, an appropriate FDR threshold should strike a balance between recapitulation of in vivo results and limiting the number of spurious binding sites. In vivo results are defined by high-confidence peaks, which are ChIP-seq peaks that were called by two peak calling programs and have a corresponding binding site sequence underlying the peak [6]. Whereas, spurious sites are accounted for by limiting the average number of HSE clusters per peak (set of potentially overlapping HSE no more than 10 bp apart from each other). Due to the repetitive nature of the HSE, a cluster is a better representative than a single site of a functional binding locus. We chose a 20% FDR threshold, which maximizes the fraction of peaks having a single HSE cluster while ensuring that a large fraction (97%) of the high-confidence in vivo peaks contain HSEs. This threshold resulted in 3735 clusters (71% with a single HSE, 20% with two HSEs overlapping by 10 bp, \sim 5% with two HSEs overlapping by 5 bp; see Figure S17).

The final set of HSE clusters was obtained by combining data from the two experimental replicates. First, a set of genomic regions was identified by intersecting the peaks from the two experimental replicates, and retaining only those peaks for which the two replicates were in close agreement (>80% of reads fall in the overlapping region). We then identified the 2896 HSE clusters that fell in these regions (\sim 77% of all clusters).

HSE cluster intensity

The problem of measuring the intensity of each peak is complicated by the fact that some peaks contain multiple, closely spaced clusters, whose contributions are difficult to disentangle. Furthermore, peaks often include trailing edges that are dominated by the background signal. To address these concerns we experimented with various measures of intensity based on the output produced by MACS (wig files giving shifted read counts in 10 bp windows) as well as the reported 'bandwidth' *B*. We considered three measures, applied to a window of radius *B* centered at each cluster: maximum read count, read count sum, and an "integrated" read count based on a biweight kernel (which produces a curve at each peak that is similar to the one implied by the peak model used by MACS). We selected the biweight kernel measure, which does the best job of handling closely spaced clusters (see Figure S18).

Computing K_d values for all genomic HSE sites

We assume that each HSE site *i* is at approximately the same initial concentration in the experiment $([HSE_i]^{initial} = C)$. Furthermore, all sites compete to bind a shared amount of free HSF, with the remaining unbound concentration denoted by [HSF]. At the end of the experiment, a fraction of site *i* is bound, with concentration $[HSE_i : HSF]$, and the remainder is unbound, with concentration $[HSE_i]$. The dissociation constant for a particular HSE site is therefore given by:

$$K_d^i = \frac{[HSF][HSE_i]}{[HSE_i : HSF]}$$

The bound HSE concentration is measured by the PB–seq experiment in terms of the number of reads at element $i(R_i)$. This leaves two unknown quantities, [HSF] and [HSE_i], in units of read counts. The first of these unknowns, [HSF], can be eliminated by considering instead the relative K_d with respect to a known reference value (for an HSE present in the experiment).

To solve for $[HSE_i]$, we express this quantity as the difference between the initial concentration C and the measured bound concentration:

$$[HSF_i] = [HSE_i]^{initial} - [HSF_i : HSF] \propto C - R_i$$

By substituting the expression for K_d^i (above) and dividing by the K_d value for the reference HSE, K_d^{ref} , we obtain an expression with a single unknown, *C*:

$$\frac{K_d^i}{K_d^{ref}} = \frac{\frac{[HSF][HSE_i]}{[HSE_i : HSF]}}{\frac{[HSF][HSE_{ref}]}{[HSE_{ref} : HSF]}} = \frac{\frac{[HSE_i]}{[HSE_{ref}] : HSF]}}{\frac{[HSE_{ref}]}{[HSE_{ref} : HSF]}} = \frac{(C-R_i)R_{ref}}{(C-R_{ref})R_i}$$

With the use of a reference dissociation value for a second HSE, we can solve for C and obtain estimates of the dissociation

constants for all other HSE sites for which read counts are available. Replacing K_d^{i} and R_i by the corresponding values for the second reference HSE and solving for *C*:

$$K_d^{ref2} = K_d^{ref2} \frac{(C - R_{ref2})R_{ref2}}{(C - R_{ref})R_{ref2}} \Leftrightarrow C = \frac{\left(K_d^{ref2} - K_d^{ref2}\right)R_{ref}R_{ref2}}{K_d^{ref2}R_{ref2} - K_d^{ref}R_{ref2}}$$

Heat Shock Element model

Our probabilistic model for HSEs was designed to capture interactions among the binding preferences of the three monomers that form the HSF homotrimer. The model consists of three PSSM-based submodels corresponding to the three 5 bp sequences (pentamers) that are bound by the HSF monomers. Each of these submodels is defined by two PSSMs, one 'strict' and one 'relaxed'. These three submodels allow for eight possible combinations of strict and relaxed pentamer binding. Within each pentamer the positions are considered independent, as in standard PSSM models.

Formally, let a candidate 15 bp HSE sequence X^{*} be composed of random variables X_{ij}^{k} where *i* is the pentamer index and *j* is the base position within that pentamer. Additionally, let each sequence have an associated unobserved random variable Y^{*} which indicates which of the eight combinations of strict/relaxed distributions are applied the corresponding X_{ij}^{k} (Figure S12). For simplicity, our model definition assumes that the middle monomer sequence has been reverse complemented and is therefore in the same orientation as the outer monomer binding sequences. We considered two versions of the model: a sparsely parameterized 'constrained' version and a more parameter-rich 'expanded' version, as described below.

Constrained version. This version of the model assumes that the three monomers share the same strict and the same relaxed PSSM-based sequence distributions. In addition, it assumes that the relaxed PSSM is defined as a more degenerate version of the strict PSSM. This is accomplished by means of a single 'degeneracy' parameter, which 'pulls' the nucleotide distribution at each position toward the uniform distribution. Specifically, the nucleotide distribution at position j of pentamer iis defined as:

$$P(X_{i,j}^{k}=b \mid Y_{i}^{k}) = \begin{cases} f_{j}^{b} & \text{if} \quad Y_{i}^{k}=strict\\ \frac{f_{j}^{b}+B}{1+4B} & \text{if} \quad Y_{i}^{k}=relaxed \end{cases}$$

where f_j^{b} is the probability of observing nucleotide *b* at position *j* of the monomer and *B* is the free parameter controlling how close to an uniform distribution the relaxed version is.

To estimate the model parameters from the HSE sequence data, we first held B fixed and then estimated the nucleotide frequencies and the prior probabilities of each strict/relaxed monomer combinations through Expectation Maximization (EM). A grid search was then used to find the value of B that maximized the model likelihood.

Estimating the model parameter updates for EM is simple for the prior but slightly more complicated for the nucleotide frequencies due to the interdependency between the strict and relaxed distributions. Nevertheless, it can be solved by using a Lagrange multiplier together with the derivatives of the expected complete log-likelihood. This produces an estimator that depends on the Lagrange multiplier and requires the use of a root finding method as part of the maximization step of EM. Figure S13C shows the results of the parameter estimation. To initialize the optimization procedure, the nucleotide frequencies were estimated from the high-confidence in vivo HSEs from [6].

Simulation study. To test the performance of this model, we estimated unconstrained strict/relaxed matrices from real data and simulated data under various distributions of Y^k . All of the parameters of the simplified model were then estimated from this simulated data, and the posterior distribution of Y^k under the model was compared with the values used for simulation (Figure S14).

Expanded version. This version of the model allows for completely separate PSSMs for the three pentamers, and completely separate strict and relaxed versions of each of these models. It has $5 \times 3 \times 2 = 90$ PSSM parameters plus seven free parameters for Y^k, for a total of 97 free parameters. Parameter estimation is again accomplished by expectation maximization, but in this case the parameter updates are trivial.

Chromatin effect and DNase I hypersensitivity models

The chromatin effect and DNase models are rule ensemble models, estimated using the RuleFit R package. This package was also used to estimate the relative importance of the model covariates. The covariates were obtained from modENCODE tracks, taking the mean value over a 200 bp window centered on the target point. Furthermore, these data were filtered to contain only points that had a value for every covariate used.

Chromatin effect model. This model estimates the ratio between the in vivo and in vitro intensities at each site from a set of chromatin covariates. Ratio values were obtained from the measured intensities using the in vitro HSE cluster coordinates. The set of HSE clusters was pre-filtered by finding a threshold on the in vitro intensities that approximately minimized the differences between experimental replicates. The threshold value, collect the mean absolute difference between experimental replicates for the ratios computed using in vitro intensities above that value; 2) compute the mean of the values collected in the previous step; 3) pick the first in vitro threshold that falls below the mean.

The model was estimated on the selected HSE clusters using different combinations of chromatin covariates. For each particular combination, an estimated Pearson correlation value was obtained from ten-fold cross validation. Furthermore, to obtain the figures presented in this paper, the data was split into 60% training data and 40% test data. The model obtained on the training data was used to make the test data predictions shown in the figures and the corresponding Pearson correlation.

DNase I hypersensitivity model. The data set used for this model was independent of the HSE clusters. 10K points were randomly sampled from across the genome with the restriction that the points did not fall within the regions shown in the figures presented in this paper or within 200 bp of the HSE cluster sites. These 10K points were used as a training set to build the model used to make the predictions for the browser tracks and at the HSE cluster regions. They were also used to estimate the Pearson correlation via ten-fold cross validation.

Prediction tracks. To produce the in vivo intensity prediction tracks, the chromatin model was applied to a version of the in vitro intensities that were scaled so that they would be comparable to the in vitro intensities. To obtain the scaling factor, we selected the top seven most accessible isolated HSE clusters (as measured by DNase hypersensitivity) that had a significant read count. The reason for these restrictions is that highly accessible

sites should be good proxies for sites that are not being influenced by chromatin effects and the sites with significant in vitro intensity should produce better estimates of the in vivo to in vitro ratio used for scaling (see Figure S4 for point choices).

The browser tracks were produced by collecting values in 50 bp steps with a 100 bp window average and applying the respective model and scaling (if needed). Values were then smoothed with a Gaussian kernel having a 100 bp bandwidth.

Supporting Information

Figure S1 HSF purification and quantification. A) Purified fulllength HSF (arrow) was estimated to be 40% pure as quantified by a silver stained gel and densitometry. B) A silver stained gel using known concentrations of BSA (10 ng/ μ l, 5 ng/ μ l, 2.5 ng/ μ l, 1.25 ng/ μ l) was used to quantify the stock concentration of purified full-length HSF (arrow) at 1.9 ng/ μ l. Note that one gel is shown, but intervening lanes were removed for simplicity. (TIF)

Figure S2 A) The mobility of the constant 200 attomole HSE probe shifts into a trimeric-HSF:HSE complex as increasing HSF is added. There is no HSF in the left-most lane, the right-most lane contains 3 nM HSF (1 nM trimeric HSF), and the intervening lanes contain two-fold serial dilutions of HSF. B) A hyperbolic curve based on the Kd equation (see Methods) was modeled using the band shift data, indicating a Kd of 97.5 pM (95% confidence interval of 59.8-158 pM). C) The constant 200 attomole HSE probe shifts into a trimeric-HSF:HSE complex as increasing HSF is added. There is no HSF in the left-most lane, the right-most lane contains 1.5 nM HSF (500 pM trimeric HSF), and the intervening lanes contain two-fold serial dilutions of HSF. D) A hyperbolic curve based on the Kd equation (see Methods) was modeled using the band shift data, indicating a Kd of 221 pM (95% confidence interval of 197-250 pM). E) This panel has the same description as panel A. F) A hyperbolic curve based on the Kd equation (see Methods) was modeled using the band shift data, indicating a Kd of 308 pM (95% confidence interval of 214-448 pM). (TIF)

Figure S3 Each panel shows smoothed 95% confidence intervals (CI) (dotted lines) for the estimated genomic Kd values (blue lines). The red and green points correspond to the Kd values determined by the EMSA assays. Error bars indicate 95% confidence intervals (CIs), as estimated in the non-linear regression (see Methods). Red points indicate those used as references to compute the genomic Kd values in each panel. The CIs shown in panels A1, B1, A2 and B2 were estimated by propagating various sources of uncertainty through our formula for estimating Kd values, using the first order Taylor expansion approximation. In panels A1 and A2, only the variance associated with the reference Kd points was considered, whereas in B1 and B2 the variance associated with the site intensity estimates was also used. At each binding site in the genome, the variance in intensity was estimated analytically from the two PB-seq replicates, after quantile normalization of the PBseq replicate intensities to remove systematic biases. In panels C1 and C2, the CIs were computed by sampling the reference Kd values from normal distributions corresponding to their respective CIs and by selecting site intensities at random from one of the two PB-seq replicate values (again after quantile normalization). To account for the uncertainty associated with the choice of reference points, we show the CIs based on the two best EMSA points in the top panels and those based on the two worst EMSA points in the bottom panels.



Figure S4 These data points (HSE cluster sites) were used to determine the scaling factor between in vivo and in vitro binding intensities in Figure 1 and Figure 3. The top left plot shows how the in vivo to in vitro intensity ratio varies with the number of points included; dashed line signals the final choice of seven points. Scatter plots show the top 30 data points (HSE cluster sites) with the highest DNase I signal, against their in vivo and in vitro intensity values; black indicates the seven chosen points. The points with higher DNase I hypersensitivity offer the best choice for unbiased scaling.

(TIF)

Figure S5 This UCSC genome browser shot provides additional examples of in vivo prediction of HSF binding intensity using chromatin and PB-seq data.

(TIF)

Figure S6 The experimentally determined ratio between in vivo ChIP-seq HSF intensity and in vitro PB–seq intensity is plotted against the predicted in vivo/actual PB–seq ratio. The Pearson correlation for each model is shown.

(TIF)

Figure S7 The bar graphs indicate the Pearson correlation of predictions versus experimental measures for each model used to predict the in vivo/in vitro binding intensity ratio (Rul: Rules Ensemble model, Lin: linear regression model). The correlations for both the training data (panel A) and the test data (panel B) are indicated.

(TIF)

Figure S8 ROC plots for in vivo HSF binding predictions. In vitro HSE sites were partitioned into bound and unbound cases by applying a threshold to the estimated in vivo intensity values. Three thresholds were considered: a permissive threshold (shown in red; 36% bound), a moderate threshold (green; 24% bound) and a strict threshold (blue; 12% bound). Each panel in the figure represents a distinct covariate set (see panel titles). For each covariate set, the corresponding rules ensemble model was applied to predict the in vivo intensity of the HSE sites. Each site was then classified as predicted to be bound or unbound by applying a threshold to these predicted intensities. These thresholds were varied to produce the Receiver Operating Characteristic (ROC) curves shown. As a baseline, we show predictions based on the scaled in vitro intensities in gray. For each ROC curve, we compute the Area Under the Curve (AUC) as a general measure of prediction performance (higher is better). Notice that the ROC curves are not highly sensitive to the threshold that is applied to the in vivo intensities, but in most cases the ensemble model produces a substantial improvement over the baseline prediction. At the same time, some covariates produce substantially better predictions than others.

(TIF)

Figure S9 For each of the models shown in Figure S6 we show the relative importance [20] of each covariate in the Rule Ensemble model built with each indicated subset of covariates to predict the in vivo/in vitro binding intensity ratio. (TIF)

Figure S10 The bar graphs illustrate the relative importance [20] of each covariate in the Rule Ensemble model built with each indicated subset of covariates to predict DNase I hypersensitivity. (TIF)

Figure S11 These bar graphs indicate the Pearson correlation of predicted versus experimentally measured DNase I sensitivity for each DNase I prediction model (Rul: Rules Ensemble model, Lin:

linear regression model). The correlations for the training data (panel A) and test data (panel B) are indicated. (TIF)

Figure S12 The structure of the HSE probabilistic sequence model recapitulates the structure of the HSE. Each hidden variable $Y_{1,}Y_{2,}Y_{3}$, determines if the respective underlying pentamer bases are drawn from a strict base distribution or a relaxed version. (TIF)

Figure \$13 Pentamers within the HSEs are dependent upon their stringency and position relative to the other pentamers. A) A composite pentamer matrix was derived from all pentamers found within PB–seq peaks. B) The strict motif from panel A and a dampening factor from panel C were used to generate a relaxed motif. C) The dampening factor was optimized to generate a relaxed motif that best explained the data. D) A probabilistic sequence model reveals that the presence of two strict and one relaxed pentamer provides the best explanation of the data. (TIF)

Figure S14 The reduced HSE sequence model predictions are compared for patterns of strict/relaxed pentamer combinations. Three different simulated patterns are shown and are recapitulated by the model.

(TIF)

Figure S15 Scatter plots show similarity of each HSE pentamer to the canonical monomer PSSM. Each point represents a PSSM estimated via MEME by sub-sampling the in vitro peaks identified by MACS. Pattern of the scatter plot shows evidence for pentamer divergence occurring on one pentamer at a time (points are spread following the axis, mainly corresponding relaxed versions of the first and second pentamers). (TIF)

Figure S16 In vivo HSF binding sites that were either detected or not detected in vitro have distinct properties. A) The composite PSSM for the 40% of HSF binding sites that are only found in vivo exhibits more degeneracy than the PSSM from the sites that are found both in vivo and in vitro. B) The binding sites exclusively found in vivo are generally more accessible, as measured by DNase I signal, than those sites found both in vivo and in vitro.

(TIF)

Figure S17 A) Balance between in vivo recall and number of per peak in vitro HSE is reached at 20% estimated FDR, corresponding to the inflection point for the number of clusters, as well as near maximal recall of high-confidence in vivo sites. B) An HSE (or HSE cluster) is considered isolated if the nearest neighbor is more than 200 bp away. An HSE (or HSE cluster) is considered overlapping if it overlaps with a single other HSE (or HSE cluster); overlaps between more than two HSE (or HSE clusters) are denoted as complex overlaps. (TIF)

Figure \$18 Three different measures were compared for computing HSE cluster intensities: max, sum and bi-weight kernel. A) Each measure was incorporated into a scatter plot of scaled intensities. Max was rejected because it produced a more compressed range of intensity values. B) In comparing the difference in intensities across replicates, the bi-weight kernel approach fares slightly better than the sum. C) The difference in magnitudes given a cluster distance on isolated clusters was compared between measures. For each distance, the isolated clusters are made to overlap an identical copy of themselves and

the magnitude difference is computed by comparing the value of the isolated cluster with the partially overlapping, using either the sum or kernel measures. Average values per intensity quartile show that bi-weight kernel measure introduces less error as a function of distance than the sum measure. D) Replicate intensities strongly correlate, as predicted, using the kernel measure.

(TIF)

References

- Field Y, Sharon E, Segal E (2011) How transcription factors identify regulatory sites in genomic sequence. Subcell Biochem 52: 193–204.
- Berger MF, Bulyk ML (2009) Universal protein-binding microarrays for the comprehensive characterization of the DNA-binding specificities of transcription factors. Nat Protoc 4: 393–411.
- Liu J, Stormo GD (2005) Combining SELEX with quantitative assays to rapidly obtain accurate models of protein-DNA interactions. Nucleic Acids Res 33: e141.
- Hesselberth JR, Chen X, Zhang Z, Sabo PJ, Sandstrom R, et al. (2009) Global mapping of protein-DNA interactions in vivo by digital genomic footprinting. Nat Methods 6: 283–289.
- Liu X, Noll DM, Lieb JD, Clarke ND (2005) DIP-chip: Rapid and accurate determination of DNA-binding specificity. Genome Res 15: 421–427.
- Guertin MJ, Lis JT (2010) Chromatin landscape dictates HSF binding to target DNA elements. PLoS Genet 6: e1001114. doi:10.1371/journal.pgen.1001114.
- Li XY, Thomas S, Sabo PJ, Eisen MB, Stamatoyannopoulos JA, et al. (2011) The role of chromatin accessibility in directing the widespread, overlapping patterns of *Drosophila* transcription factor binding. Genome Biol 12: R34.
- Kaplan T, Li XY, Sabo PJ, Thomas S, Stamatoyannopoulos JA, et al. (2011) Quantitative models of the mechanisms that control genome-wide patterns of transcription factor binding during early *Drosophila* development. PLoS Genet 7: e1001290. doi:10.1371/journal.pgen.1001290.
- Narlikar L, Gordan R, Hartemink AJ (2007) A nucleosome-guided map of transcription factor binding sites in yeast. PLoS Comput Biol 3: e215. doi:10.1371/journal.pcbi.0030215.
- Pique-Regi R, Degner JF, Pai AA, Gaffney DJ, Gilad Y, et al. (2011) Accurate inference of transcription factor binding from DNA sequence and chromatin accessibility data. Genome Res 21: 447–455.
- Boyle AP, Song L, Lee BK, London D, Keefe D, et al. (2011) High-resolution genome-wide in vivo footprinting of diverse transcription factors in human cells. Genome Res 21: 456–464.
- Guertin MJ, Petesch SJ, Zobeck KL, Min IM, Lis JT (2011) Drosophila heat shock system as a general model to investigate transcriptional regulation. Cold Spring Harb Symp Quant Biol.
- Hayashida N, Fujimoto M, Tan K, Prakasam R, Shinkawa T, et al. (2010) Heat shock factor 1 ameliorates proteotoxicity in cooperation with the transcription factor NFAT. EMBO J 29: 3459–3469.
- Gonsalves SE, Moses AM, Razak Z, Robert F, Westwood JT (2011) Wholegenome analysis reveals that active heat shock factor binding sites are mostly associated with non-heat shock genes in *Drosophila melanogaster*. PLoS ONE 6: e15934. doi:10.1371/journal.pone.0015934.
- Pepke S, Wold B, Mortazavi A (2009) Computation for ChIP-seq and RNA-seq studies. Nat Methods 6: S22–32.
- Chen R, Snyder M (2010) Yeast proteomics and protein microarrays. J Proteomics 73: 2147–2157.
- 17. Gordan R, Hartemink AJ, Bulyk ML (2009) Distinguishing direct versus indirect transcription factor-DNA interactions. Genome Res 19: 2090–2100.
- Kharchenko PV, Alekseyenko AA, Schwartz YB, Minoda A, Riddle NC, et al. (2011) Comprehensive analysis of the chromatin landscape in *Drosophila* melanogaster. Nature 471: 480–485.
- Gilchrist DA, Dos Santos G, Fargo DC, Xie B, Gao Y, et al. (2010) Pausing of RNA polymerase II disrupts DNA-specified nucleosome organization to enable precise gene regulation. Cell 143: 540–551.
- Friedman JH, Popescu BE (2008) Predictive learning via rule ensembles. Ann Appl Stat 2: 916–954.
- Lee H, Kraus KW, Wolfner MF, Lis JT (1992) DNA sequence requirements for generating paused polymerase at the start of *Hsp70*. Genes Dev 6: 284–295.

Table S1ModENCODE identification number or GEOaccession number for each data set used in the paper.(XLS)

Author Contributions

Conceived and designed the experiments: MJG JTL. Performed the experiments: MJG. Analyzed the data: MJG ALM. Wrote the paper: MJG ALM AS JTL. Developed the statistical models: ALM AS.

- John S, Sabo PJ, Thurman RE, Sung MH, Biddie SC, et al. (2011) Chromatin accessibility pre-determines glucocorticoid receptor binding patterns. Nat Genet 43: 264–268.
- Robertson AG, Bilenky M, Tam A, Zhao Y, Zeng T, et al. (2008) Genome-wide relationship between histone H3 lysine 4 mono- and tri-methylation and transcription factor binding. Genome Res 18: 1906–1917.
- Wu W, Cheng Y, Keller CA, Ernst J, Kumar SA, et al. (2011) Dynamics of the epigenetic landscape during erythroid differentiation after GATA1 restoration. Genome Res 21: 1659–1671.
- Tursun B, Patel T, Kratsios P, Hobert O (2011) Direct conversion of C. elegans germ cells into specific neuron types. Science 331: 304–308.
- Biddie SC, John S, Sabo PJ, Thurman RE, Johnson TA, et al. (2011) Transcription factor AP1 potentiates chromatin accessibility and glucocorticoid receptor binding. Mol Cell 43: 145–155.
- von Hippel PH, Revzin A, Gross CA, Wang AC (1974) Non-specific DNA binding of genome regulating proteins as a biological control mechanism: I. the lac operon: Equilibrium aspects. Proc Natl Acad Sci U S A 71: 4808–4812.
- Lin S, Riggs AD (1975) The general affinity of lac repressor for *E. coli* DNA: Implications for gene regulation in procaryotes and eucaryotes. Cell 4: 107–111.
- Fritsch M, Wu C (1999) Phosphorylation of Drosophila heat shock transcription factor. Cell Stress Chaperones 4: 102–117.
- Voss TC, Schiltz RL, Sung MH, Yen PM, Stamatoyannopoulos JA, et al. (2011) Dynamic exchange at regulatory elements during chromatin remodeling underlies assisted loading mechanism. Cell 146: 544–554.
- He HH, Meyer CA, Shin H, Bailey ST, Wei G, et al. (2010) Nucleosome dynamics define transcriptional enhancers. Nat Genet 42: 343–347.
- Hu G, Schones DE, Cui K, Ybarra R, Northrup D, et al. (2011) Regulation of nucleosome landscape and transcription factor targeting at tissue-specific enhancers by BRG1. Genome Res 21: 1650–1658.
- John S, Sabo PJ, Johnson TA, Sung MH, Biddie SC, et al. (2008) Interaction of the glucocorticoid receptor with the chromatin landscape. Mol Cell 29: 611–624.
- Sharon E, Lubliner S, Segal E (2008) A feature-based approach to modeling protein-DNA interactions. PLoS Comput Biol 4: e1000154. doi:10.1371/ journal.pcbi.1000154.
- He X, Chen CC, Hong F, Fang F, Sinha S, et al. (2009) A biophysical model for analysis of transcription factor interaction and binding site arrangement from genome-wide binding data. PLoS ONE 4: e8155. doi:10.1371/journal.pone. 0008155.
- Enoki Y, Sakurai H (2011) Diversity in DNA recognition by heat shock transcription factors (HSFs) from model organisms. FEBS Lett 585: 1293–1298.
- Sakurai H, Takemori Y (2007) Interaction between heat shock transcription factors (HSFs) and divergent binding sequences: Binding specificities of yeast HSFs and human HSF1. J Biol Chem 282: 13334–13341.
- Tao H, Liu W, Simmons BN, Harris HK, Cox TC, et al. (2010) Purifying natively folded proteins from inclusion bodies using sarkosyl, triton X-100, and CHAPS. BioTechniques 48: 61–64.
- Li H, Durbin R (2009) Fast and accurate short read alignment with burrowswheeler transform. Bioinformatics 25: 1754–1760.
- Barrett T, Troup DB, Wilhite SE, Ledoux P, Rudnev D, et al. (2009) NCBI GEO: Archive for high-throughput functional genomic data. Nucleic Acids Res 37: D885–90.
- Zhang Y, Liu T, Meyer CA, Ecckhoute J, Johnson DS, et al. (2008) Model-based analysis of ChIP-seq (MACS). Genome Biol 9: R137.
- Bailey TL, Williams N, Misleh C, Li WW (2006) MEME: Discovering and analyzing DNA and protein sequence motifs. Nucleic Acids Res 34: W369–73.